Author's response to reviews

Title: Patient- and delivery-system factors related to acceptance of HIV counseling and testing services among TB patients in South Africa: A qualitative study with community health workers and program managers

Authors:

J. Christo Heunis (heunisj.hum@ufs.ac.za)
Edwin Wouters (edwin.wouters@ua.ac.be)
Wynne E Norton (wenorton@uab.edu)
Michelle C Engelbrecht (engelmc@ufs.ac.za)
Gladys N Kigozi (kigozign@ufs.ac.za)
Anjali Sharma (anjali.sharma@downstate.edu)
Camille Ragin (camille.ragin@downstate.edu)

Version: 7 Date: 21 February 2011

Author's response to reviews: see over
Dear Dr. Kent:

Thank you for the opportunity to finally edit and resubmit our manuscript.

We have edited the manuscript as follows in response to the requests received from the Independent Editorial Production Team:

1. Title: We notice that the title on the submission system is different to the title on the manuscript title page. Please ensure the two are identical.

The two titles are now identical.

2. Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is ‘First name Initial Last name’ (e.g. Joe F Bloggs).

We checked that this was correct and in the preferred style.

3. Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

The affiliations are now in the format and includes the information required.

4. Abstract: please ensure that the Abstract in the manuscript matches exactly the Abstract in the submission system.

The two abstracts are now identical.

5. References: Please make sure all references are cited and appear in numerical order.

All references are cited and in numerical order.

6. References: please provide a functioning link to the website in reference 20.
A functioning link for reference 20 has now been provided.

7. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions.

Typographical errors have been corrected and final revisions have been made.

Please let me know if any of our responses require additional clarification or if there are any other issues that we need to address.

Thank you.

Sincerely,

Christo Heunis, Ph.D.
Associate Professor
Centre for Health Systems Research & Development
Faculty of the Humanities
University of the Free State
205 Nelson Mandela Drive
Bloemfontein, South Africa, 9300
heunisj@ufs.ac.za